Author's response to reviews

Title: Quality of interaction between primary health-care providers and patients with type 2 diabetes in Oman: An observational study

Authors:

Nadia Abdulhadi (nadia.abdulhadi@ki.se)
Mohammed Ali Al-Shafee (shafae4@omantel.net.om)
Claes-Goran Ostenson (claes.ostenson@karolinska.se)
Asa Vernby (asa.vernby@ki.se)
Rolf Wahlstrom (rolf.wahlstrom@ki.se)

Version: 4 Date: 24 November 2006

Author's response to reviews: see over
Quality of interaction between primary health-care providers and patients with type 2 diabetes in Oman: An observational study
Journal: BMC Family Practice
MS : 2019791415106325

Formatting changes:

Qualifications – We removed the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system.

Affiliations - Each affiliation was written in full, in the following format: department, institute, city, and country.

The superscript 5 was removed from the author list and 1 was used for Asa Vernby, and also we removed the five from the affiliations.

Contact person – Was renamed the ‘Corresponding author' who is the first author. In addition, the first author mail address and telephone no. were added.

Running title – Was removed from the title page.

Word count – was removed from the manuscript.

References - We used square brackets rather than rounded brackets for the citations in the text.

The Abbreviations section was moved to before the Competing interests.